Author's response to reviews

Title: Differing mental health practice among general practitioners, private psychiatrists and public psychiatrists

Authors:

Nadia Younes (nadiayounes@wanadoo.fr)
Marie-Christine Hardy-Bayle (MCHardyBayle@ch-versailles.fr)
Bruno Falissard (falissard_b@wanadoo.fr)
Viviane Kovess (vkovess@mgen.fr)
Marie-Pierre Chaillot (MCHAILET@ch-versailles.fr)
Isabelle Gasquet (isabelle.gasquet@sap.aphp.fr)

Version: 5 Date: 18 September 2005

Author's response to reviews:

I've added the French version of the Mental Health Practice Questionnaire as asked.
Best regards
Nadia Younes MD